Author's response to reviews

Title: The impact of a computerized decision aids on vaginal versus cesarean section delivery - Study protocol for a randomized controlled trial

Authors:

Saeid Eslami (EslamiS@mums.ac.ir)
Azam Aslani (azamaslani@yahoo.com)
Fatemeh Tara (taraf@mums.ac.ir)
Leila ghalihi (leila.ghalichi@gmail.com)
Fatemeh Erfanian (ErfanianF@mums.ac.ir)
Ameen Abu-Hanna (a.abu-hanna@amc.uva.nl)

Version: 2 Date: 20 July 2014

Author's response to reviews:

According preliminary review:
1. We removed the conflict of interest statement from your title page.
2. We included the date your study was registered with your trial registration number at the end of your Abstract.
3. We included a list of abbreviations used and their meanings, after your Trial Status.
4. We included an Authors’ Contributions section at the end of the manuscript, before the reference list.
5. We included an Acknowledgement